Author's response to reviews

Title: Population normative data for the 10/66 Dementia Research Group cognitive test battery from Latin America, India and China: a cross-sectional survey

Authors:

Ana L Sosa (drasosa@hotmail.com)
Emiliano Albanese (emiliano.albanese@iop.kcl.ac.uk)
Martin Prince (martin.prince@iop.kcl.ac.uk)
Daisy Acosta (daisyacosta@codetel.net.do)
Cleusa P Ferri (cleusa.ferri@iop.kcl.ac.uk)
Mariella Guerra (mariella.guerra.1066@googlemail.com)
Yueqin Huang (dengy@mail.tsinghua.edu.cn)
K S Jacob (ksjacob@cmcvellore.ac.in)
Juan Llibre de Rodriguez (j.llibre-rodriguez@iop.kcl.ac.uk)
Aquiles Salas (aquiles@cantv.net)
Fang Yang (Fang.Yang.1066@googlemail.com)
Ciro Gaona (ciro.gaona.1066@googlemail.com)
Amuthavalli Thiagarajan Joteeshwaran (kkvh9991@iop.kcl.ac.uk)
Guillermina Rodriguez (guillermina.rodriguez.1066@googlemail.com)
Gabriela Rojas de la Torre (gabriella.delatorre.1066@googlemail.com)
Joseph D Williams (jdwilliams.1066@googlemail.com)
Robert Stewart (r.stewart@iop.kcl.ac.uk)

Version: 3 Date: 23 July 2009

Author's response to reviews: see over
Response to Reviewers’ comments

We are very grateful for the Reviewer’s further helpful comments and are pleased to respond as follows:

Abstract – We had misunderstood the original comment as to which WLM/WLR reference was being highlighted. We have identified the correct instance which is indeed a typo, and have corrected this. We apologise for the oversight.

Table 6 – The typo in question is the mean score for this column and row which we have now corrected. The ‘-2.5’ statistic is correct.

Formatting revisions:

1. We have added a section documenting informed consent and details of the ethics committees which reviewed this study.

2. We have altered the text of the Authors’ Contributions section as requested.